Author’s response to reviews

Title: Asian-Specific Total Knee System: 5-14 year Follow-up Study

Authors:

Kunihiro Hosaka (kunihiro619@hotmail.com)
Shu Saito (saitou.syu@nihon-u.ac.jp)
Takao Ishii (ishii.takao@nihon-u.ac.jp)
Sei Mori (moriseimori@gmail.com)
Takanobu Sumino (sumino.takanobu@nihon-u.ac.jp)
Yasuaki Tokuhashi (tokuhashi.yasuaki@nihon-u.ac.jp)

Version: 2 Date: 14 June 2011

Author’s response to reviews:

I changed 'All patients provided informed consent to participate in this study' to 'All patients provided informed consent to participate in this study which was conducted in accordance with the Helsinki Declaration.'